Author's response to reviews

Title: Nonossifying fibroma presenting aneurysmal bone cyst: a case report

Authors:

Akio Sakamoto (akio@med.kyushu-u.ac.jp)
Takeaki Ishii (takeaki@surgpath.med.kyushu-u.ac.jp)
Yoshinao Oda (oda@surgpath.med.kyushu-u.ac.jp)
Yukihide Iwamoto (yiwamoto@ortho.med.kyushu-u.ac.jp)

Version: 2 Date: 23 August 2012

Author's response to reviews: see over
Dear Editors

Enclosed please find a manuscript which we wish to submit for publication in “Journal of Medical Case Reports” under the category of “Case Reports”. We have changed the style of the manuscript in order to meet the journal. We transfer all copyright ownership of the manuscript entitled “Nonossifying fibroma presenting aneurysmal bone cyst: a case report” by Akio Sakamoto, et al. We warrant that article is original, is not under consideration by any other journal, and has not been previously published. Your editorialists have our full permission to carry out any necessary linguistic adjustments deemed to be required.

Your kind consideration of our manuscript would be greatly appreciated.

Sincerely yours,

Akio Sakamoto, M.D., Ph.D.